Author’s response to reviews

Title: The impact of feedback during formative testing on study behaviour and performance of (bio)medical students: a randomised controlled study

Authors:

Dick Thijssen (dick.thijssen@radboudumc.nl)

Maria Hopman (maria.hopman@radboudumc.nl)

Marielle van Wijngaarden (m.vanwijngaarden@radboudumc.nl)

Joos Hoenderop (joost.hoenderop@radboudumc.nl)

rene bindels (rene.bindels@radboudumc.nl)

Thijs Eijsvogels (thijs.eijsvogels@radboudumc.nl)

Version: 1 Date: 27 Aug 2018

Author’s response to reviews:

Dear Prof. Sultan Ayoub Meo,

Thank you very much for providing the opportunity to change our manuscript and respond to your comments.

We believe we have addressed all concerns, which we have shortly commented on below.

We hope our manuscript has improved as is potentially acceptable for publication in your journal.

Yours sincerely,

Dick Thijssen
RESPONSE TO COMMENTS (OUR RESPONSE IN CAPITALS)

Technical Comments:

1. Title page – Please upload the title page at the front of your manuscript file.

WE HAVE MADE THE SUGGESTED CHANGES.

2. Please rename Introduction to Background section.

WE HAVE MADE THE SUGGESTED CHANGES.

3. Please note that all manuscripts must contain all the following sections under the heading 'Declarations': Abbreviations, Ethics approval and consent to participate, Consent for publication, Availability of data and material, Competing interests, Funding Authors' contributions, Acknowledgements

WE HAVE MADE THE SUGGESTED CHANGES.

4. Please include a statement on consent to participate in the “Ethics approval and consent to participate” section of the Declarations, and state whether consent was written or verbal, together with an explanation if it was verbal.

WE HAVE MADE THE SUGGESTED CHANGES.

Editor's comments:

26% of similarity index, to modify the manuscript, minimize the similarity index mainly. I have highlighted the high similarity in the following sections. Introduction section (Page 2, line 1-9);
WE HAVE MADE THE SUGGESTED CHANGES AND REWROTE THESE SECTIONS TO MINIMISE SIMILARITY.